VC- /Week 42 starting 16 October 06
SM18 Operation Meeting (Tuesdays) 17 Oct 06

Bench status  on sunday 15-oct-2006 at: 
10/15/2006 9:21:48 PM  
	3
	A1
	MBBR2382
	ICS 4 Final connection
	297.15

	9
	A2
	MBBL1380
	Prep 4 TRU Connection
	1.90

	8
	B1
	SSS518
	Cryo problem
	5.41

	4
	B2
	SSS353
	PT 13.2 Resist. Meas.
	297.41

	1
	C1
	SSS278
	Prep 3 CDW1 HV
	296.52

	2
	C2
	SSS339
	ICS 6 Disconnect Magnet WP07
	296.43

	11
	D1
	SSS520
	PT5.4: Powering Corrector EF
	1.89

	12
	D2
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	296.34

	6
	E1
	MBBR2389
	PT 13.2 Resist. Meas.
	301.35

	5
	E2
	MBBL2357
	PT 13.2 Resist. Meas.
	296.88

	7
	F1
	MBBL1409
	PT 6.3 Training 3
	10.18

	10
	F2
	MBBL1408
	Cryo problem
	296.88


  
CONFIG :  D 6 - AS 4 -  IRSSS 2 ;   
Supervisor:  stephane ;Eq support : 16-3602 
14 Magnet Tests completed ( 5 Magnet Stripped ) 
08 Dipoles tests
()

5 Arc-SSS  tests (317 had long MM)
1  500-series (515 despite  ROBTC but did not reach 12KA in 2 quenches but…)
 0  600-series [ ] 
General Points:
· SSS339 & Q-loc not recorded in SMTMS
· Situation from Cryogenics:  J-P Lamboy or ACR Rep.
· Situation SSSS  :D1 & D2 issues J Halik 
· A.O.B..
	Magnets under Test from 08-10-2006 to 14-10-2006 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBAL2336 
	[image: image3.png]



	


	05-10-2006 
	41 h.
	19 h.
	20 h.
	16 h.
	12 h.
	09-10-2006 
	108 h.
	Stripping
	
	

	2
	TBA1
	MBBL1373 
	[image: image5.png]



	


	10-10-2006 
	40 h.
	22 h.
	17 h.
	16 h.
	16 h.
	14-10-2006 
	111 h.
	Stripping
	
	

	3
	TBA2
	MBBR2390 
	[image: image7.png]



	


	03-10-2006 
	40 h.
	19 h.
	19 h.
	19 h.
	10 h.
	08-10-2006 
	107 h.
	Stripping
	
	

	4
	TBA2
	MBBR2393 
	[image: image9.png]



	


	08-10-2006 
	32 h.
	22 h.
	21 h.
	15 h.
	6 h.
	12-10-2006 
	96 h.
	Stripping
	
	

	5
	TBB1
	SSS145 
	[image: image11.png]



	


	06-10-2006 
	34 h.
	13 h.
	23 h.
	9 h.
	28 h.
	11-10-2006 
	107 h.
	Stripping
	
	

	6
	TBB2
	SSS354 
	[image: image13.png]



	


	07-10-2006 
	0 h.
	9 h.
	11 h.
	11 h.
	30 h.
	10-10-2006 
	61 h.
	Stripping
	
	

	7
	TBC1
	SSS358 
	[image: image15.png]



	


	06-10-2006 
	30 h.
	17 h.
	17 h.
	8 h.
	11 h.
	09-10-2006 
	83 h.
	Stripping
	
	

	8
	TBC1
	SSS360 
	[image: image17.png]



	


	09-10-2006 
	39 h.
	20 h.
	18 h.
	8 h.
	37 h.
	14-10-2006 
	122 h.
	Stand By
	
	

	9
	TBD1
	SSS317 
	[image: image19.png]



	


	29-09-2006 
	88 h.
	16 h.
	178 h.
	8 h.
	32 h.
	13-10-2006 
	322 h.
	Stripping
	
	

	10
	TBD2
	SSS515 
	[image: image21.png]



	


	07-10-2006 
	0 h.
	16 h.
	13 h.
	12 h.
	26 h.
	10-10-2006 
	67 h.
	Stand By
	
	

	11
	TBE1
	MBBR2387 
	[image: image23.png]



	


	06-10-2006 
	22 h.
	25 h.
	23 h.
	19 h.
	25 h.
	11-10-2006 
	114 h.
	Stripping
	
	

	12
	TBE2
	MBBR2388 
	[image: image25.png]



	


	04-10-2006 
	53 h.
	21 h.
	20 h.
	18 h.
	34 h.
	10-10-2006 
	146 h.
	Stripping
	
	

	13
	TBF1
	MBBR2381 
	[image: image27.png]



	


	07-10-2006 
	39 h.
	28 h.
	16 h.
	16 h.
	23 h.
	12-10-2006 
	122 h.
	Stripping
	
	

	14
	TBF2
	MBBR2374 
	[image: image29.png]



	


	07-10-2006 
	73 h.
	19 h.
	15 h.
	17 h.
	26 h.
	13-10-2006 
	150 h.
	Stripping
	
	

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAL2336
	11759 A.
(D1_U)
	12327.6 A.
(D2_L)
	12586.3 A.
(D1_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL1373
	12005.3 A.
(D1_L)
	12501.4 A.
(D1_L)
	12666.8 A.
(D1_L)
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	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBR2390
	10884 A.
(D1_U)
	12383.9 A.
(D2_L)
	12850 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBR2393
	11366.1 A.
(D1_L)
	12407.1 A.
(D1_U)
	12807.8 A.
(D2_L)
	12850 A.
[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	SSS145
	12368.7 A.
(QE_12)
	12789.5 A.
(QE_12)
	12850 A.
[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS354
	11704.6 A.
(QI_12)
	11533.4 A.
(No Data)
	12379.3 A.
(QI_34)
	[image: image36.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS358
	11058.5 A.
(QE_12)
	11449.6 A.
(QE_12)
	12633.4 A.
(QE_12)
	12724.9 A.
(QE_12)
	[image: image37.png]



	-
	-
	-
	-
	-
	-
	-
	-

	8
	SSS360
	11607.5 A.
(QI_12)
	12698.2 A.
(QI_12)
	12816.2 A.
(QE_12)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	SSS317
	10921.6 A.
(QI_12)
	12561.5 A.
(QE_34)
	12740.9 A.
(QE_34)
	12846 A.
(QE-12)
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-
	-

	10
	SSS515
	11309.8 A.
(QI_12)
	11776.3 A.
(QI_34)
	12167.3 A.
(QI_12)
	[image: image40.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	11
	MBBR2387
	12671.4 A.
(D2_L)
	12850 A.
[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	12
	MBBR2388
	12136.9 A.
(D2_U)
	12850 A.
[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	13
	MBBR2381
	11470.2 A.
(D2_L)
	12820 A.
(D1_U)
	12850 A.
[image: image43.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	14
	MBBR2374
	11048.5 A.
(D1_L)
	12082.7 A.
(D2_U)
	12411 A.
(D2_L)
	[image: image44.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
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   LONG MM on SSS317

 

	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuedsay_10_N
	TBC2
	SSS339 : After the first training quench alarms in AQA, signals not coherent.
	MTM
	AQA
	Giorgio contacted. Problem coming from the changes induced by the shaft localization (channels shifted). Giorgio advised to create a dummy file  without Q_loc and use it to analyse the quench. Giorgio corrected the configuration Wednesday morning.

	Wednesday_11_M
	TBD1
	SSS317 : During special LHC cycle, cryo OK lost, communication problem in ACR supervision
	ACR
	
	Test re-launched.

	Wednesday_11_M
	TBA2
	MB2393: No quench heater data was available after warm-up quench (training quench a03).
	MTM
	
	Guy advised to re-launch the warm-up quench.

	Thursday_12_A
	TBA2
	MB1380: Scanning of the magnet, pointer of the mobic not found.
	
	
	

	Saturday_14_A
	TBF2
	MB1408: Temperature anti-cryostats too low.
	ACR
	
	ACR team asked to ignore the alarm.

	Sunday_15_M
	TBB1 and TBF1
	SSS518, MB1409: Temperature on TBB1 and TBF1 was stuck at 4.5K.
	ACR
	
	Pumping system is not working properly.

	Sunday_15_M
	TBF 
	MB1408, MB1409: Significant temperature difference was observed between TT821 and TT147. 
	MTM
	
	Guy contacted.

	Sunday_15_A
	TBB1
	SSS518: TCl_aux fault.
	ACR
	
	Sensor to be changed.. SSS warmed up, naturally sensor changed Monday and SSS cooled down again.

	Sunday_15_A
	TBF2
	MB1408: Big leak found in the magnet at warm. Vacuum broken by cryo. 

Problem with anti-cryostat.
	MTM
	
	Action ?

Anti-cryostat cable was damaged in the aperture 1. OP team replaced this cable by one from A cluster. Salem has been informed.

	Sunday_15_N
	TBA2
	MB1380: Q_loc position different in the TRU between the two apertures. Switch adapted accordingly.
	MTM
	
	

	Sunday_15_N
	TBF1
	MB1409.: Training quench,  lf crate hanging.
	AB/CO
	lf
	Lf crate re-started.

	Monday_16_M
	TBE1,TBA2
	EDMS upgrade: HWREC stopped. MM2 launching in MB1380 and upload of WP07on MB2389 delayed.
	IT
	EDMS
	

	Monday_16_M
	TBA2
	MB1380: MM2 checked by Stéphane. Two sectors in each aperture displaying wrong magnetic field
	MTM
	
	Stéphane asked Guy to check the measurement chain (Amplifiers, integrators.).

	Monday_16_M
	TBD1
	SSS320: MQTL powering, the magnet did not reach 600 A after 9 ramp.
	MTM
	
	Stéphane asked to continue the training up 550 A adding some quenches because magnetic measurements is required on this magnet. The MQTL reached 534 and then de-trained. Stéphane asked for 2 last quenches and stop. He will change the MM accordingly.

	Monday_16_A
	TBA2
	MB1380: At the end of MM3 (~1000 A) the current failed down. Slow PA, lost of cryo OK?

Same problem happened during the test ramp
	MTM
	
	Stéphane validated the measurement.

Giorgio informed.

	
	
	
	
	
	

































































